Author's response to reviews

Title: Polymorphism of the FABP2 gene: a population frequency analysis and an association study with Cardiovascular Risk Markers in Argentina.

Authors:

Laura L C Gomez (lgomez@fcm.uncu.edu.ar)
Sebastian S M Real (sreal@unsl.edu.ar)
Marta M S Ojeda (msojeda@unsl.edu.ar)
Sergio S Gimenez (sergio.gimenez@osep.mendoza.gov.ar)
Luis L S Mayorga (lmayorga@fcm.uncu.edu.ar)
Maria M Roque (mroque@fcm.uncu.edu.ar)

Version: 5  Date: 19 June 2007

Author's response to reviews: see over
we have included the following formatting changes:
Author list - We have written the names of the authors in the manuscript exactly as they are in the submission system.

Title - we removed the full stop from the end of the title.

Authors' contributions - we indicate that all authors read and approved the final manuscript.

Website addresses - we have removed websites from the text, and cited from the main reference list only in the following structure: Website title [www.website.com].

Figures-> Figure legends. Underlining - We did removed the underlining of any section headings.